Author's response to reviews

Title: The association between temporal changes in the use of obstetrical intervention and small-for-gestational age live births

Authors:

Amy Metcalfe (amy.metcalfe@albertahealthservices.ca)
Sarka Lisonkova (slisonkova@cw.bc.ca)
KS Joseph (ksjoseph@cfri.ca)

Version: 4 Date: 31 August 2015

Author's response to reviews:

Q: Please confirm whether the NCHS data used is publicly available or whether permission was necessary for accessing it. If so, please state who granted permission to use this data

A: This data is publicly available. We have added a statement to our methods confirming this and have provided the website where this data can be obtained. Ethics approval was granted from the Research Ethics Board at the University of British Columbia. We have added a sentence to the methods section confirming this.

Q: Please remove the running head, word count, number of figures, tables and appendices, abbreviations and Financial support section from the title page.

A: Done

Q: Please provide a Conclusions section at the end of the Discussion

A: Done

Q: Please place the List of abbreviations after the Conclusions section

A: Done

Q: Please provide after the list of abbreviations a Competing Interests, Author's Contributions and Acknowledgements sections, formatted according to our Instructions for Authors

A: Done

Q: Please remove all figures from the main manuscript, as these have already been correctly uploaded as separate figure files

A: Done